NIHR National Institute for Health Research

Research for Patient Benefit Final Report Form

Project Title	Investigating Factors prEdictive of and associatEd with psychologicaL WELLbeing in Transgender People following Gender Affirming Treatment (GAT). (FEEL-WELL)	
Reference Number	PB-PG-0817-20003	
Contracting Organisation	Nottinghamshire Healthcare NHS Foundation Trust	
Approved Duration	20	
Current Duration	20	
Contracted Start Date	01/06/2019	
Contracted End Date	30/01/2021	
Original Award	149,982.00	
Current Award	149,982.00	

Project Details

Grant Title:	Investigating Factors prEdictive of and associatEd with psychologicaL WELLbeing in Transgender People following Gender Affirming Treatment (GAT). (FEEL-WELL)		
Reference Number:	PB-PG-0817-20003	Contracting Organisation:	Nottinghamshire Healthcare NHS Foundation Trust
Contracted Start Date:	20	Current Duration:	20
First Contracted Start	01 June 2019	Contracted End Date:	30 January 2021
Original Award:	149 982 00	Current Award:	149,982.00
Original Award.	149,902.00		
Full Name Organisation Email Address	Pro No Joi	ofessor Jon Arcelus ttinghamshire Healthcare Ni n.arcelus@nottingham.ac.uk	HS Foundation Trust

Research Team

Chief Investigator	Professor Jon Arcelus (Nottinghamshire Healthcare NHS Foundation Trust)
--------------------	---

Joint Lead Applicant:	

Selection List	Professor Jon Arcelus		
Role in research	Co-ordinating the project		
Selection List	Dr Boliang Guo		
Role in research	Statistics		
Selection List	Dr Elena Nixon		
Role in research	Quantitative methodology		
Selection List	Dr Walter Pierre Bouman		
Role in research	Clinical input		
Selection List	Mr Jamie Pallas		
Role in research	PPI and links with Community		
Selection List	Professor Andrew Yip		
Role in research	Qualitative methodology		
Selection List	Mr Jonathan Lee		
Role in research	Finance		

Involvement of NIHR Infrastructure

Please indicate which NIHR Infrastructure organisations were involved in your research.

Please describe the role of each organisation in your research

None of the above infrastructures have been involved in this project, a part from advise directly from RfPB NIHR

Changes to Research Team

Please outline any changes that have been made to the research team over the course of the research, including an explanation of why they were required.

Jamie Pallas left the team and he was substituted by Carla English, PPI representative and working in the same organisation as Jamie Pallas

Scientific Summary

Please provide a structured summary of your work.

Background: Transgender people are those whose gender identity does not match the sex assigned at birth based on their sexual characteristics. There is strong evidence of high levels of mental health problems and poor quality of life among transgender people before they start Gender affirming treatment (GAT), (cross-sex hormones therapy (CHT) and/or surgery). The role of GAT in mental health symptoms and whether those symptoms are predictors of treatment outcome is unknown. Factors that affect wellbeing post treatment have not been explored.

Aims and objective: To explore factors associated with and predictive of transgender people's psychological wellbeing following GAT.

Methods: The project have consisted in 3 studies. The first two studies focus on transgender people who initiated GAT 18 months previously using a quantitative longitudinal design to analyse already collected data longitudinally. The 3rd study focusses on people who initiated GAT more than 5 years ago using a qualitative methodology.

Key findings:

In the first study data from 178 trans Participants was analysed. The results showed that GAT reduced mental health symptomatology significantly for depression (P < .001) and non-significantly for anxiety (P = .37). Social support predicted reduction in depression.

For the second study data from 137 people was analysed. Results showed that mental health symptoms pretreatment did not predict life satisfaction post treatment.

The qualitative studies identified six themes. These themes included; work and finance, the positive impact of transition, social support, experiences of media and social media, healthcare, and societal acceptance. Each of the themes identifies a factor which participants highlighted as impacting either positively or negatively on their wellbeing.

Output

The results have been presented in 2 International conferences (WPATH Virtual conference 2020, and EPATH 2021 conference), and National conferences (BAGIS virtual conference 2020). An article has been published in an open access journal (IF of 3.35) and the paper has been used for the development of the next edition of the Global guidelines for trans health (Standards of Care 8th Edition by the World Professional Association for Transgender Health). Two publications to be submitted. The team worked with Gendered Intelligence to develop documents from the results of the study.

Conclusions

1- Gender Affirming Hormone treatment reduces mental health symptoms, particularly Depression. This is the largest longitudinal study showing the evidence of hormone treatment in mental health.

2- levels of anxiety and depression pre-treatment do not predict life satisfaction post hormone treatment, however gender congruence in associated with life satisfaction- suggesting the importance of treatment and that mental health symptoms (anxiety or depression) should not be a barrier for treatment

3- The importance of education to health professionals on trans issues as this is important for wellbeing

4- The damage that social medial and negative media representation can do to the wellbeing of trans people.

Future Plans

- To explore the routes of anxiety post treatment more in depth and develop interventions of this.
- To explore the barrier for treatment and work towards reduction of waiting times in gender services (currently 3 years)

Keyword 1:	mental health		
Keyword 2:	Trans health		
Keyword 3:	Wellbeing		
Keyword 4:	Depression		
Keyword 5:	Anxiety		
			
Keyword 6:	Gender affirming hormone treatment		
Keyword 7:			
Keyword 8:			

Plain English Summary

Please provide a plain English summary of your research.

Aims and objective: To explore which factors are associated with transgender people's psychological wellbeing

Background: Transgender people are those whose gender identity does not match their sex assigned at birth based on their sexual characteristics. There is little information as to how hormones affect people's mental health symptoms and which factors are associated with wellbeing after hormone treatment.

Methods: The project consisted of 3 studies. The first two studies follow trans people on hormones for 18 months and the last study interviews trans people to find out factors associated to their wellbeing.

Key findings:

1- Hormones reduce mental health symptoms, particularly Depression. This is the largest study showing the evidence of hormone treatment in mental health.

2- Anxiety and depression do not predict life satisfaction post hormone treatment, suggesting that mental health symptoms (anxiety or depression) should not be a barrier for treatment

3- The importance of education to health professionals on trans issues as this is important for wellbeing

4- The damage that social medial and negative media representation can do to the wellbeing of trans people.

Dissemination and Output

Presented in 2 International conferences and 1 National conference. Article published which has been used for the development of the next edition of the Global guidelines for trans health. Two other publications are about to submitted. The team worked with Gendered Intelligence to develop easy to read documents from the results of the study.

PPI

The team has been working closely with the community through Gendered Intelligence particularly as part of the interview study and the dissemination.

Future Plans

- To explore why anxiety does not reduce as much as depression with hormones and what can be done about this
- To explore barriers for treatment and work towards reduction of waiting times in gender services (currently 3 years)

Please tick the box if this section of the report has been written with members of the public who have been involved in the research.

Confirmed

Aims and Objectives

Please describe the original aims and objectives of the research.

The main objective of the project is to explore factors influencing psychological wellbeing (focussing on mental health and life satisfaction) of transgender people following GAT. The study will focus on two groups of transgender people: 1) those who initiated GAT 18 months previously (to analyse predictors) and 2) those who initiated GAT 5 years previously (to investigate associated factors). The aims of each study are:

1. Study 1: To investigate changes in mental health symptoms (depression and anxiety) following 18 months of GAHT

2. Study 2: To investigate the predictive role of specific factors such as socio-demographics, interpersonal (e.g. social support) and mental health symptoms collected before GAT, on mental health problems and life satisfaction 18 months from the initiation of GAT.

3. Study 3: To explore, using a qualitative design, factors influencing psychological wellbeing of transgender people who initiated GAT 5 years previously; such factors may be: attributed to the presence of specific life events/experiences in participants' lives which may have led to low levels of well-being; associated with intra-/inter-personal difficulties; and/or lack of social support, to name but a few.

Changes to Aims and Objectives

If the aims and objectives changed, please explain in what way and why.

no changes

Description of Research

Please provide a structured summary of your work.

Background

Cross-sectional studies show that transgender people are more likely than cisgender people to experience depression and anxiety before gender-affirming hormone treatment (GAHT). However, the effect of GAHT on mental health in transgender people, and the role of other factors that may have a predictive effect, is poorly explored. Additionally, in comparison to the general cisgender population trans people have been identified by previous research as having a poorer quality of life (QoL). While prior research has focused on the wellbeing of transgender people prior to initiating Gender Affirming Medial Treatment (GAMT) or 12-18 months after initiation of GAMT, little qualitative research has been carried out to examine the factors that impact transgender peoples life satisfaction and wellbeing after this stage. The aim of these studies are;

1) To use a longitudinal methodology to investigate the effect of 18-month GAHT on depression and anxiety symptomatology and the predictors on mental health outcomes in a large population of transgender people

2) To investigate factors that impacted upon the wellbeing of transgender people who had initiated GAMT 5 or more years ago using semi-structured interviews.

<u>Methods</u>

For the *quantitative studies*, data already collected was used. Participants (n = 178) completed a sociodemographic questionnaire, the Hospital Anxiety and Depression Scale (HADS), the Multidimensional Scale of Perceived Social Support (MSPSS) and the Autism Spectrum Quotient—Short Version (AQ-Short) at preassessment (T0) and at 18 months after initiation of GAHT (T1). In addition, the Gender congruence and life satisfaction scale (GCLS) was used as outcome measure at T1.

For the *qualitative study* data was collected using semi-structured interviews and analysed through grounded theory and thematic analysis. The analysis focused on identifying factors discussed in the interviews that impacted upon the wellbeing of the participants. A group (n=23) of eligible participants were recruited through social media and transgender support

organizations. Participation eligibility criteria was being 18 years old or over and having initiated GAMT five or more years ago.

<u>Findings</u>

The quantitative studies found that from T0 to T1, symptomatology was significantly decreased for depression (P < .001) and non-significantly reduced for anxiety (P = .37). Scores on the MSPSS predicted reduction in depression, while scores on the AQ-Short predicted reduction in anxiety. A multiple regression with five predictor variables was conducted to explore the predictors of Gender Congruence and life satisfaction at T1 separately. The predictors for these regressions were age, assigned sex at birth, MSPSS score, HADS-A score, and HADS-D score at T0. It was found that MSPPS scores were found to be predictors to Gender Congruence. In addition, the multiple regression analysis for life satisfaction was not significant (F(5,114)=1.380,p=0.237) with an explained variance of life satisfaction scores of 5.7% ($R^2 = .057$). None of the predictors included in the regression were identified as a significant predictor of cluster 2 and for life satisfaction scores.

The qualitative study identified six themes and 18 sub-themes that participants highlighted as impacting on their wellbeing. Some of these themes were the same as factors prior research has found in cisgender populations (while being viewed through the lens of trans experience), while other themes were more specific to the trans experience.

Each of the themes identifies a factor which participants highlighted as impacting either positively or negatively on their wellbeing. The themes identified are split into several sub themes in which more detailed analysis of the factors can be carried out. The themes identified include;

<u>Theme 1</u> – experiences of healthcare (Sub themes: 1a- Positive and negative experiences of trans health services, 1b- The right non-specialist support): Participants discussed positive experiences from support received from medical systems and professionals. The data also highlighted the negative impact that a lack of support,

understanding, and communication could have both in regards to trans healthcare services and non-specialist healthcare services (such as GPs).

This theme is of course shaped by the trans experience of the participants as they have gone through the process of accessing and acquiring GAMT, a process that can be taxing and difficult. When these experiences were positive, participants tended to frame them in terms where a good relationship of respect and understanding had been built up between them and their clinician, while negative experiences tended to be due to a lack of communication and understanding. While this does initially seem to be an obvious point, this process for many treatment seeking trans people is an extremely personal and stressful one and a lack of understanding from clinicians as well as a lack of clear and regular communication from the clinics themselves can only make the experience more difficult. The early discharge from trans health services was a factor of dissatisfaction among some and something that services should be reviewing. As a consequence of this early discharge, many trans people need to be looked after by non-specialist services, who have a lack of understanding of their specific health needs. Specific health needs for trans people later on in transition include monitoring of hormone levels, surgical interventions (and post-surgical care) for those who want them. Some participants discussed that non-gender specialist medical professionals lacked sufficient knowledge of trans issues and the process in which trans people could access gender treatment. They described how this impacted on their GAMT even at this stage of their transition, perhaps even more so due to the increased separation from contact with GICs. This emphasises the importance of both clear and regular communication between GICs, patients, and non-specialist medical professionals, as well as the importance of training for non-gender specialist medical professionals on the issues that trans people may experience when interacting with the healthcare systems and professionals. This lack of specific health-related support was an important factor for many, which requires further research.

<u>Theme 2</u> – Positive impact of transition (Sub themes: 2a- Increasing resilience and confidence, 2b- Reducing Misgendering): Participants discussed various positive impacts of transitioning on their current well-being. These impacts were both in regard to improvements in personal resilience and also in terms of reducing misgendering by others.

<u>Theme 3</u> – Societal Acceptance (Sub themes: 3a- Fitting in with the trans community, 3b- Socialising and passing 3b.1- Challenges with passing 3b.2- Socialising and socialisation as a trans person, 3c- negotiating coming out, 3d- promoting societal acceptance): Participants discussed how their well-being was impacted by others' reactions to their identity and how they presented. Participants further discussed how these reactions shaped their experiences and approaches both in terms of other trans people and with cisgender people.

<u>Theme 4</u> – Social support (Sub themes: 4a- Having the right social support, 4b- Losing support through transition, 4c- Feeling lonely or isolated, 4d- Support from the LGBT+ community): As expected, relationships were found to have variable impact upon the participants' well-being. With some relationships aiding in improving well-being while a lack of relationships or support from these relationships often negatively impacted participants causing feelings of isolation or loneliness. Of note in the participants responses were the relationships with LGBTQ+ people and more specifically other trans peers.

Participants discussed the importance of understanding experiences when receiving social support. This was most prevalent when participants were discussing receiving social support from the LGBTQ+ community and from their trans peers. The shared experiences of these groups allowed for participants to feel more at ease and able to discuss issues that they were experiencing without the additional emotional labour of explaining the nuances of the experience to those who had not experienced something similar. This however was not a universal experience. Interestingly, some participants experienced issues even in trans spaces, this was particularly the case for participants who were gender diverse. One participant had described the usefulness of trans and LGBTQ+ spaces however also highlighted the negative impact that some trans people's expectations of other trans bodies had on their well-being.

<u>Theme 5</u>- Impact of media and social media (Sub themes: 5a- Negative representation of trans issues by media, 5b- Experience of social media): Representation of trans people in the media and trans people's experience of social media can often be negative. This negative representation and experiences were highlighted by participants as factors impacting their life-satisfaction. Social media and the media in general were identified as having a significant detrimental impact for many. This was due to the negative coverage of trans issues by the media which may be more prevalent in the UK, and not universal. This impact was not only detrimental to participants well-being but also changed their media consumption habits, leading them to be more cautious about the media sources they consumed to avoid those sources which are transphobic. The additional stress of avoiding what can be prevalent transphobic media sources may also have an impact on trans people's wellbeing, this can also involve the experience of distress associated with learning that previously though safe media sources are no longer safe. In addition to this, there was a discussion around social media in which it was highlighted that online harassment is still a significant issue for many trans people when they are open about their experiences.

<u>Theme 6</u> - Influence of work and financial situations (Sub themes: 6a- Job satisfaction, 6b- Workplace support, 6c-Financial security, 6d- Living environment): As seen in research in cis people, work and financial situation was linked to wellbeing. Participants highlighted several aspects which impacted their well-being ranging from positive factors such as getting satisfaction from their work and being supported in the workplace to negative factors which focused around housing pressures and a lack of financial or employment security. While many of these factors are similar to those found in a cisgender population there are some aspects which were specific to those who have trans experience such as the presence or lack of protective legislations in the workplace and hostile living environments due to transitioning.

Conclusions

The quantitative studies show that GAHT reduces symptoms of depression which are predicted by having higher levels of social support. Although anxiety symptoms also reduce, the changes are not significant and high levels of anxiety still remain post-GAHT. These results highlight the important mental health benefits of GAHT. Support services (professional, third sector or peer support) aiming at increasing social support for transgender individuals should be made available. In addition, the lack of associations found between pre-treatment mental health symptoms and life satisfaction post treatment indicate that these factors should not constitute as barriers to access to GAMT for treatment-seeking transgender people.

A common narrative through many of the themes developed in the qualitative study is that of the experience of support. This narrative presents itself differently across the various themes. This is illustrated by participants discussing the importance of more institutional supports such as knowledgeable non-specialist medical professionals and workplace protections, as well as discussing the value of social support from family and friends and the impact of its loss.

The conclusions from the results of the qualitative study highlight the importance of the positive impact of third-party support organizations, protective legislations, awareness in the general public, and the provision of recommendations and guidance for specialist and non-specialist healthcare providers. These factors could be used for guidance in how best to provide additional services and care for transgender people who are undergoing GAMT. It is important to consider that these positive influences are already afforded to many cisgender people.

Intellectual Property, Commercialisation and Clinical Adoption

Please provide brief details of IP outputs arising from this research.

There are no issues related in this section.

Actual and Anticipated Impact

Please provide a brief impact statement.

The results have been presented in 2 International conferences (WPATH Virtual conference 2020, and EPATH 2021 conference), and a National conference (BAGIS virtual conference 2020).

An article has been published in an open access journal (IF of 3.35). The paper has been used in the next edition of the Global guidelines for trans health (Standards of Care 8th Edition by the World Professional Association for Transgender Health).

Two publications to be submitted.

The team worked with Gendered Intelligence, to develop easy to read documents from the results of the study which has been distributed among their members.

Describe the impact the research has already achieved or might achieve in the short, medium and long term.

The results of the quantitative studies have been used as part of the development of the 8th edition of the Standards of Care for professionals working with trans people. The Standards of care are the Global guidelines for trans professionals which include, surgeons, endocrinologist, psychologist etc. They are developed by the World Professional Association for Transgender Health (WPATH).

The guidelines are developed using the best evidence-based information possible, hence the results of our studies (1 and 2) will be used for this guideline. This is important as our study is the largest longitudinal study available showing the benefits of gender affirming hormone treatment in mental health and showing that mental health symptoms pre-treatment do not predict life satisfaction post treatment.

This evidence is vital in highlighting the benefit of treatment and the importance of trans health services.

It is expected that the 3 studies will have an impact in the way professionals work with trans people around the world. And the recommendations made in education (part of the 3rd study) of health professionals.

Dissemination

Please describe how you have disseminated your research findings and what your plans for further dissemination are.

The 3 studies have been written for publications in peer reviewed journals. The first study was published in the "andrology" journal as an open access publication and it was highlighted in "nature" reviews urology. The second and 3rd study are been developed to be submitted for publication in May or June 2021.

The 3 studies have been presented separately in several local, national and international conferences, which include;

- Research conference SURAGth- Nottingham 2019
- Research conference Division of Psychiatry- University of Nottingham 2020
- Annual conference of British Association of Gender Identity Specialist (BAGIS)- Virtual 2020
- World Professional Association for Transgender Health- Virtual Conference- International- November 2020
- European Professional Association for Transgender Health- accepted- August 2021

A summary of the results in lay language and using graphics (see attached example) was developed with the charity Gendered Intelligence, which was posted in their website, twitted and send to NHS England.

Publications

Number published	2
Number in press	0
Number submitted	1
Number in preparation	1

Publications and Other Outputs

Please add any research outputs that incorporate findings from the research and have been published since the last progress report.

Where outputs have not been published online (or the full text is not available through Europe PubMed Central (Europe PMC) or open access from the publisher) please append a copy of the final version as an annex to this report in the **'supporting documentation'** section.

Output Title	Long term effect of gender affirming hormone treatment on depression and anxiety symptoms in transgender people: A prospective cohort study
--------------	---

Was this output submitted to CCF?

Type of Output

Is the output available in PubMed?

Article

Yes

No

PubMed ID (PMID) for output

https://pubmed.ncbi.nlm.nih.gov/32777129/

Central? Yes

If the output makes any recommendations for policy and/or clinical practice, please provide details

The article is being used for the development for the global guidelines for trans health specialist developed by the World Professional Association for Transgender Health (WPATH) https://www.wpath.org/

The global guidelines are called the Standards of Care and they are currently developing the 8th edition where thanks to this article recommendations will be made of the medical necessity of hormone treatment for trans people. https://www.wpath.org/publications/soc

Is this output published open access?

Yes

How could this output be utilised? (Please tick the one option	Informing health and social care policy
that is most applicable):	and/or practice

Date output

10/08/2020

appeared/published:			
Output Title Feel-Well Project			
Was this output submitte	d to CCF?		No
Type of Output	Presentation		
Is the output available in PubMed? No			No
Place where the output a	opeared e.g. iou	rnal or conference	
Conference			
If the output makes any re	ecommendation	s for policy and/or clinic	al practice, please provide details
Conference presentation at	the PPI meeting	in Nottingham	
If available online please provide a hyperlink:			
Please give the venue: University of Nottingham			
How could this output be utilised? (Please tick the one option that is most applicable):			
Date output appeared/published:	11/07/2019		
Output Title		What factors influence the life satisfaction and wellbeing of transgender people who have initiated Gender Affirming Medical Treatment 5 or more years ago?	
Was this output submitte	d to CCF?		No
Type of Output Presentation			
Is the output available in PubMed?			No
Conference			
If the output makes any recommendations for policy and/or clinical practice, please provide details			
Conference presentation at the International conference of the World Professional Association of Transgender Health- virtual conference due to Covid			
If available online please provide a hyperlink:			

Please give the venue:		Virtual conference			
How could this output be that is most applicable):	utilised? (Pleas	e tick the one option	Informing health and social care policy and/or practice		
Date output appeared/published:	06/11/2020				
Output Title		Long term effect of gence depression and anxiety prospective cohort study	ler affirming hormone treatment on symptoms in transgender people: a		
Was this output submitte	d to CCF?		No		
Type of Output	Presentation				
Is the output available in	PubMed?		No		
Place where the output ap	opeared e.g. jou	rnal or conference			
Conference					
If the output makes any re	ecommendation	s for policy and/or clinic	cal practice, please provide details		
Conference presentation at of the study for the develop	the World Profestment of the Stand	ssional Association for Tra dards of Care- Global guid	ansgender Health discussing the implications deline 8th edition.		
lf available online please hyperlink:	provide a				
Please give the venue:		Virtual			
How could this output be that is most applicable):	utilised? (Pleas	e tick the one option	Informing health and social care policy and/or practice		
Date output appeared/published:	06/11/2020				
Output Title		Predictors of gender affirming hormone treatment in transgender people: Data Mental health and life satisfaction outcomes from on a longitudinal study.			
Was this output submitte	d to CCF?		No		
Type of Output	Presentation				
Is the output available in	PubMed?		No		
Place where the output a	opeared e.g. jou	rnal or conference			
Conference					
Reference: PB-PG-0817-	Date submitted:				

If the output makes any recommendations for policy and/or clinical practice, please provide details							
Discussing the role of mental health workers in trans health, this abstract has been accepted as a presentation at the European Professional Association for Transgender Health- Gothenburg- Sweden 2021							
		1					
If available online please provide a hyperlink:							
Please give the venue:		Gothenburg- Sweden					
How could this output be that is most applicable):	utilised? (Pleas	e tick the one option	Informing further research				
Date output appeared/published:	05/08/2021						
Output Title		Gender-affirming therapy	y linked to mental health				
Was this output submittee	d to CCF?		No				
Type of Output	Article						
Is the output available in I	PubMed?		Yes				
PubMed ID (PMID) for output https://www.nature.com/articles/s41585-020-00377-6							
Is the full text of this outp Central?	ut available on	Europe PubMed	Yes				
If the output makes any re	ecommendation	s for policy and/or clinic	cal practice, please provide details				
A research highlight was wr	ritten based on o	ur previous publication in	Nature				
Is this output published o	pen access?		No				
How could this output be that is most applicable):	utilised? (Pleas	Informing further research					
Date output appeared/published:	10/09/2020						
Please detail any awards and/or prizes received by the team as a result of undertaking the research.							
The study that has been published has been submitted for an award to the Institute of Mental Health at the University of Nottingham, we are awaiting the outcome of it.							

Patient and Public Involvement

Please provide a summary of the patient and public involvement in this research.

Aim: to provide guidance in development, participant recruitment, and dissemination of the research

Methods: A representative of Gendered Intelligence has been working as co-applicant from the beginning of this project. They have been involved in the development of the project, and analysis of the results and writing up of some of the publications. The PPI have been involved in recruitment for the 3rd project and as a focus group for developing the questions to be used in this study (as the 1st and 2nd project used already collected data). The PPI have been instrumental in development of dissemination.

Results: Recruitment during the 3rd study was quick and easy using Gendered Intelligence's access to a wide network of potential participants. The PPI guidance in the development of the interview guide highlighted aspects that may have not been considered without them. The PPI focused dissemination of this project has aided in the consideration of alternative methods of dissemination and PPI. These methods included clear lay language summaries using graphics to get across the key points and outcomes of the research, along with a more in depth document made up of abstract-like summaries of the three studies that make up this project.

Discussion and conclusions: The project was improved through the use of PPI as it facilitated various aspects of the research. These aspects included the design process of the separate studies where trans voices and experiences could guide the development of research materials such as the interview guide, recruitment where engaging with Gendered intelligence and the community both on social media and through community groups facilitated participant recruitment, and dissemination of the research in which Gendered intelligence have been instrumental in aiding in developing lay language summaries of the research for public dissemination.

Reflective / critical perspective: This project highlighted the importance of gathering clinical data with funded research to reduce logistical issues within data collection and collation.

An important step in this area of research is to aim for a more inclusive sample, including more non-binary participants particularly in the longitudinal quantitative studies. The quantitative studies were unable to include these participants due to a lack of non-binary participants. While the qualitative study did include some non-binary participants it would have benefited from a larger number of these participants to provide a more representative sample.

This research project benefited greatly from the inclusion of PPI at all stages of the process with Gendered Intelligence and the community in general being able to provide guidance in the

design process of individual studies, materials for these studies and dissemination options for the results. This study also benefited from the use of a trans researcher as part of the research team who as a member of the community themselves were able to provide knowledge and advice based on lived trans experience. The researchers trans status also provided benefits in the interview section of the project where participants may have been put at ease talking to a researcher who had similar lived experiences.

The use of quantitative and qualitative methods in this project was also of significant benefit as it provided a deeper insight into the lived experiences of trans people through the qualitative data which was complimented by the more generalizable large-scale qualitative data.

Please tick the box if this section of the report has been written with members of the public who have been involved in the research.

Confirmed

Future Research Plans

Please outline your next steps to maximise patient benefit or to further inform policy development/evaluation.

The study aimed to provide preliminary data regarding wellbeing in trans people who have been on gender affirming hormone treatment for long time. The qualitative study provides important information regarding the vital part that education for professionals makes in their wellbeing. Future studies will be developed to explore the educational needs for health professionals regarding trans people and how, improving training and education, can improve trans people's wellbeing.

The study also found the damaging effect of social media in people. The charity that we work with uses social media to inform trans people and we will be working with them to explore how trans people can be safe when using social media, this is primarily important for young trans people.

The 1st study indicated that although depressive symptoms reduce following treatment, anxiety does not reduce much. This is an important finding and suggest the need to offer targeted interventions. We have already started a pilot study looking at adapting existing interventions for trans people with the aim to evaluate this new intervention.

The studies also found that mental health symptoms pre-treatment do not predict life satisfaction following treatment. Having life satisfaction as primary outcome is important in trans health as that is the primary aim of the intervention. In addition, the studies also found that hormone treatment were vital in reducing mental health symptoms, particularly depression. In view of the high levels of suicidality in this population, particularly in young people, interventions aiming at cutting waiting times for treatment are vital. In view of this, the team has already started developing a project aiming at exploring ways to reduce waiting times in gender services, which currently are over 3 years.

Publication of Research Findings

Please indicate if there is any information that you do not wish us to place in the public domain and explain why.

There is no information that we don't want to be published. Our aim is to publish the 3 studies in peer reviewed journals.

Data Sharing

Where applicable, please provide a statement about your data sharing and accessibility. It should provide a clear and positive indication:

- Where and when the data will be shared
- Who can access the data
- How the data can be obtained

For the 1st and 2nd study, data already obtained from patients from the Nottingham Centre for Transgender Health was used. This data belongs to Nottinghamshire Healthcare NHS Foundation Trust. Data could be shared with others in the future following an application to the Nottingham centre for Transgender Health and an approval by Nottinghamshire Health Care NHS Foundation Trust. This will also require an amendment to the Ethics application.

Data for the 3rd study will be kept safe at Nottingham University. Ethics approval by the University was provided with the understanding that only the research team was able to access the data, if others outside the research team wish to use this data, with the agreement with the research team, they will need to apply for an amendment to the ethics application.

Post-Award Monitoring

Please provide the details of the individual whom we can contact for post-award monitoring of this project. Usually this will be the Chief Investigator, however, another individual, for example a project manager, may be named instead.

Contact name	Prof. Jon Arcelus			
	r			
Contact address	Institute of Mental Health, Jubilee campus, University of Nottingham, Nottingham NG7 2TU			
Contact telephone number	01158230417			
Contact email address	jon.arcelus@nottingham.ac.uk			

This report is independent research funded by the National Institute for Health Research. The views expressed in this publication are those of the author(s) and not necessarily those of the NHS, the National Institute for Health Research or the Department of Health and Social Care.

Supporting Documentation

PROFESSOR JON ARCELUS (Orcid ID : 0000-0002-3805-0180)

Article type : Original Article

Long term effect of gender affirming hormone treatment on depression and anxiety symptoms in transgender people: A prospective cohort study.

Zoë Aldridge¹, Shireen Patel¹, Boliang Guo¹, Elena Nixon¹, Walter Pierre Bouman^{1,2}, Gemma L. Witcomb³, and Jon Arcelus^{1,2}.

¹Institute of Mental Health, Faculty of Medicine and Health Sciences, University of Nottingham, Nottingham, United Kingdom.

²Nottingham Centre for Transgender Health, Nottingham, United Kingdom.

³School of Sport, Exercise and Health Sciences, Loughborough University, Loughborough, United Kingdom.

Short title: Effect of hormones in trans people's mental health

Keywords: hormone therapy, transgender, longitudinal, mental health, social support, autism

Correspondence: Professor Jon Arcelus. Institute of Mental Health, Jubilee Campus, Triumph Road, University of Nottingham, Nottingham, NG7 2TU, UK. Email: Jon.Arcelus@nottingham.ac.uk. Tel: 44 (0) 115 82 30417

Jon Arcelus. ORCID iD. https://orcid.org/0000-0002-3805-0180.

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the <u>Version of Record</u>. Please cite this article as <u>doi: 10.1111/ANDR.12884</u>

Abstract

Background

Cross-sectional studies show that transgender people are more likely than cisgender people to experience depression and anxiety before Gender Affirming Hormone Treatment (GAHT). However, the effect of GAHT on mental health in transgender people, and the role of other factors that may have a predictive effect, is poorly explored.

Objectives

Using a longitudinal methodology, this study investigated the effect of 18 months GAHT on depression and anxiety symptomatology and the predictors on mental health outcomes in a large population of transgender people.

Materials and Methods

Participants (n=178) completed a socio-demographic questionnaire, the Hospital Anxiety and Depression Scale (HADS), the Multidimensional Scale of Perceived Social Support (MSPSS) and the Autism Spectrum Quotient Short Version (AQ-short) at pre-assessment (T0) and at 18 months after initiation of GAHT (T1).

Results

From T0 to T1, symptomatology was significantly decreased for depression (P <0.001) and non-significantly reduced for anxiety (P=0.37). Scores on the MSPSS predicted reduction in depression, while scores on the AQ-short predicted reduction in anxiety.

Discussion

GAHT reduces symptoms of depression which are predicted by having higher levels of social support. Although anxiety symptoms also reduce the changes are not significant and high levels of anxiety still remain post GAHT.

Conclusions

These results highlight the important mental health benefits of GAHT. Support services (professional, third sector or peer-support) aiming at increasing social support for transgender individuals should be made available.

Treatment-seeking transgender people who are not on hormone treatment, have reported high levels of mental health problems, particularly anxiety, depression and self-harm, which are likely caused by a number of internal and external stressors [1-12]. Studies examining mental health in transgender people have primarily focused on individuals attending transgender health services and hence those who are likely to experience a higher level of distress about their assigned sex at birth. These studies have primarily looked crosssectionally at levels of anxiety [7, 13, 14], depression [7, 15-17] and self-harm [18-20].

With regards to anxiety, several studies have demonstrated high levels in transgender people before Gender Affirming Hormone Treatment (GAHT) [21.22]. For example, Bouman et al. [13] found that levels of anxiety in transgender people were three times higher than those in a matched sample from the general population. This study also found that transgender males were more anxious than transgender females. Interestingly, the high scores on autistic traits found among this population have been suggested to be a product of the high levels of anxiety and low self-esteem often experienced by this group [15] and not autism per se [23]. However, a recent study has demonstrated stability in autistic traits following GAHT [24].

Similar to anxiety, high levels of depression have also been reported in transgender individuals, prior to GAHT [16, 18, 22]. Witcomb et al. [16] reported that transgender people prior to receiving GAHT had a four-fold increased risk of a probable depressive disorder compared to a matched control sample from the general population. Why this is the case is unclear, but social factors such as lack of general social support [9, 25-27], parental support [28], and peer support have been found to be associated with depressive symptoms among transgender people [29,30]. Experiences of transphobic discrimination are associated with increased odds of suffering with depression [31] independent of other types of discrimination, e.g., racism. This suggests that transgender people who are ethnic minorities are at even greater risk, due to the intersectional experience of discriminatory events. In addition, while unemployment increases the risk of depression in the general population [32] and transgender people have been found to have a higher unemployment rate than cisgender people [33] – being in employment is associated with higher levels of experienced transphobia and fear of disclosing mental health problems in the transgender population [34, 35].

Another factor that has been associated with mental health problems in treatment-seeking transgender people is age. Younger transgender people report high levels of bullying [36] and very high levels of self-harm [19, 37], which have been associated with increased anxiety as well as effects on self-esteem, family relationships and social life, which all negatively influence mental wellbeing.

While these studies have provided valuable insight, the use of cross-sectional methodologies to examine the impact of the above factors, particularly the role of GAHT in mental health is limited. Therefore, it is critical to explore this on a within-subject basis using a longitudinal design. This is the most effective approach to show the effects of GAHT on mental health as it provides the opportunity to examine individuals prior to and during GAHT.

A small number of longitudinal studies that focus on the effect of GAHT on mental health do exist. Colizzi et al. [38] reported significant reductions in mental health symptoms after the initiation of GAHT with anxiety reducing from 50% to 17% and depression from 24% to 11%. Heylens et al. [39] also showed significant reductions in symptoms of anxiety and depression after the initiation of GAHT to the point where they resemble those of the general population. These studies are however not without limitations. Heylens et al.'s[39] study has a small sample size (n=57), while Colizzi et al.'s [38] study is limited by the lack of evaluation of factors that may have impacted on the mental health of their participants, such as social support. Both studies describe the need to replicate their findings. In contrast, Bränström and Pachankis [40] using the Swedish population register showed no significant association between the likelihood of accessing mental health treatment and time since initiation of GAHT. The limitation of their study includes primarily that accessing mental health services does not necessarily reflect actual mental health and there is little additional information about the type of mental health treatment received by their participants. These limitations mean that this study cannot provide reliable evidence regarding the role of GAHT

on the mental health symptoms of transgender people, and this information is vital in order to provide an evidence base of GAHT improving overall quality of life of transgender people.

While the available longitudinal studies have provided valuable evidence of the effect of GAHT on transgender people's mental health there is a requirement to replicate these studies addressing their limitations. With this in mind, the primary aim of this study is to examine the effect of GAHT on anxiety and depression symptoms. The study will focus on those who have been on treatment for over 18 months as this allows for enough time for GAHT to produce physical, bodily changes but before surgical procedures have taken place, which could bias the results. As some physical changes can be quicker in assigned females at birth than in assigned males at birth (e.g., voice change with testosterone) [2], which can affect mental health outcome following GAHT, the results of GAHT in anxiety and depression for both groups will be presented separately. It is hypothesized that an improvement in mental health will take place in those assigned male and female at birth following GAHT treatment. Unfortunately, due to the long waiting list for gender affirming surgical treatment in the United Kingdom (UK), it is unlikely that people will have undergone these interventions before this time. The secondary aim of this study is to examine pre-GAHT factors which may be predicting changes in anxiety and depression following GAHT. The predictors selected for this study are based on the literature and include ethnicity, age, assigned sex at birth, civil status, employment, social support, and autistic traits. This study hypothesised that symptoms of depression and anxiety would be significantly decreased after 18 months of GAHT.

Participants

Participants were invited to take part through a national transgender health service in Nottingham, UK. This service is part of the National Health Service (NHS) and offers assessment for suitability of GAHT as well as chest and genital reconstructive surgery. The service also offers GAHT and speech and language therapy. The service accepts referrals from people aged 17 and over who are seeking, or considering, medical transition.

Procedures

The sample consisted of individuals who attended an assessment at the transgender health service from November 2014 to March 2018, who agreed participation, and who were not on GAHT prior to the assessment. Prior to the clinical assessment every patient was invited to participate in the study. If agreed, they were invited to complete a baseline questionnaire pack (T0). The pack included a socio-demographics questionnaire (age, sex assigned at birth, gender identity, ethnicity, employment status, relationship status, and whether participants were taking cross-sex hormones and/or blockers pre-assessment - as a significant proportion young people are referred from the only existing child and adolescent transgender health services in the United Kingdom). Validated questionnaires regarding anxiety and depression (HADS), social support (MSPSS), and autistic traits (AQ-Short) were also included in the information pack. Data was only included if participants returned a signed consent form with the study questionnaires.

Participants who consented and returned TO questionnaires were invited to complete a T1 questionnaire 18 months after commencing GAHT. The T1 questionnaire pack consisted of a HADS questionnaire. This allowed a comparison of changes in depression and anxiety symptoms before and after GAHT. Data was collected in October 2019. Except for the data analysis the study was primarily unfunded and set up in a busy clinic.

The Hospital Anxiety and Depression Scale (HADS)[41]

The HADS is a 14-item self-report screening scale originally developed to indicate the possible presence of anxiety and depression states in medical nonpsychiatric outpatient clinics. The HADS consists of two subscales, HADS-Depression (HADS-D) and HADS- Anxiety (HADS-A). Each sub-scale has seven items that are rated on a 4-point Likert scale that ranges from 0-4 with some items reverse scored. A maximum total of 21 can be obtained on each sub-scale. A score of 0-7 on both scales implies a non-clinical range, whilst a score of 8-10 suggests the possible presence of a depressive or anxiety disorder. A score of 11 or higher suggests the probable presence of a depressive or anxiety disorder. Caseness of depression and anxiety has been suggested for scores above 8 [42]. The HADS has previously been used with transgender individuals [13, 43]. For depression (HADS-D) this gave a specificity of 0.7 and a sensitivity of 0.9. For anxiety (HADS-A) this gave a specificity of 0.78 and a sensitivity of 0.9. In this study, the Cronbach's alpha for depression was 0.76 and for anxiety 0.68.

The Multidimensional Scale of Perceived Social Support (MSPSS) [44]

The MSPSS is a 12-item self-report scale to record levels of social support from family, friends and significant others. The measure consists of three subscales to measure the three different types of support. Items are rated on a Likert scale that ranges from 1 ("very strongly agree") to 7 ("very strongly disagree"). To calculate subscale scores, items from each subscale are added together and divided by 4. A total score is calculated by adding together all 12 items and dividing by 12. The mean and total scores range from 1 to 7 with a higher score indicating a higher level of perceived social support. A mean total scale score ranging from 1 to 2.9 can be considered low support; a score of 3 to 5 can be considered moderate support and a score from 5.1 to 7 can be considered high support. The MSPSS has previously been used with transgender individuals [26, 45]. In this study, the Cronbach's alpha was 0.89.

Autism Spectrum Quotient Short Version (AQ-Short) [46]

The AQ-Short is a 28-item self-report questionnaire designed to measure autistic traits to give an indication of where the person lies on the continuum of the spectrum, ranging from

This article is protected by copyright. All rights reserved

Tools

healthy to autistic [47]. It is a shortened version of the validated AQ-50 [48]. It consists of two higher order factors related to autistic traits, including numbers and patterns (which assesses the extent to which people are fascinated by numbers, dates, patterns and categories) and social behaviours. The AQ-short is a 4-point Likert scale ranging from "definitely agree" to "definitely disagree", with some items reverse scored. Total scores range between 28 and 112. The AQ-Short has previously been used with transgender populations [23, 24, 49]. Higher scores represent higher levels of autistic traits. Although not intended to be a diagnostic tool, a cut-off of \geq 70 was found to have a sensitivity of 0.94 and specificity of 0.91 to discriminate between an autism sample and a community sample. Cronbach's alpha was 0.86.

Data analysis

Data analyses were performed using the Statistical Software Package Stata 16 [50]. Stata 16 was used to conduct power analysis. Only those participants not on GAHT at assessment (T0) were included in the regression analysis. All missingness were imputed using analytical model with 20 imputed datasets generated for each model. Paired sample t-tests were used to determine if there had been a significant change in the HADS-D and HADS-A subscales from T0 to T1. Multiple regression was conducted to explore ethnicity, employment status, relationship status, age, assigned sex, MSPSS, and AQ. The hypothesis regarding whether the specific factors were predictive of changes in anxiety and depression was tested via a moderator analysis, entering only the subscales found to be significant in the linear regressions, and a product of their combined centred scores. This was tested via a multiple regression. Bonferroni corrections were used to correct multiplicity issue if needed. Although data are not normally distributed Allison [51]) states that normality is the least important assumption of regression and since data met the assumptions for linearity, homoscedasticiy, and absence of multi-colinearity or extreme outliers, a multiple regression analyses was conducted. The socio-demographic categories were split into two distinct groups for each category as seen in Table 1. Assigned sex at birth instead of gender identity was used in the socio-demographics in view of the many different gendered identities described, as an analysis based on gender identities would have made the analyses too complex to interpret; this followed previous studies approaches [23, 24]. The MSPSS and AQ-28 Short at T0 were significant factors predicting change in HADS-D and HADS-A at T1.

To check the robustness of regression estimates sensitive to missingness, all regression models were re-run on observed only data and the results were examined against the results from imputed dataset.

Ethical approval for the study was received from the NHS Ethics Committee (14/EM/0092) and the Research and Development Department at Nottinghamshire Healthcare NHS Foundation Trust in line with Health Research Authority guidance [52], which included approval for individuals aged 17 and over to sign giving their consent without the need for additional parental consent.

Results

Socio-demographic characteristics of the participants

A total of 1,271 participants were assessed between November 2014 and March 2018, completed T0 questionnaires and agreed participation in the study. Seventy-one percent (N=906) could be included in the analysis as they had not received hormones prior to assessment. Of these, 178 (20%) went on to complete a T1 questionnaire after 18 months of GAHT, indicating a response rate of 20%. Responders did not differ from non-responders in terms of either demographic characteristics or baseline AQ-Short scores, but they were significantly less anxious at baseline than non-responders (median 9 vs 8, p=0.001; z=3.225) (see Table 1).

The age range from the 178 participants that completed T0 and T1 questionnaires ranged from 17-79 years with a median age of 23 years. More than half of the participants (n=95; 53.3%) were assigned male sex at birth and 83 (46.7%) were assigned female sex at birth. The large majority of participants classified themselves as white (n=167; 94%), were single (n=120; 69%) and were in employment (n= 75; 41%) or students (n = 58; 32%). Participants who were assigned male sex at birth were more likely to be in employment compared to participants assigned female sex at birth (47% vs 33%) while more participants assigned female sex at birth were of assessment (78% vs 57%) and a higher percentage of participants assigned male sex at birth were divorced/separated (18% vs 2%) (see Table 1).

Please insert Table 1 around here

Anxiety and depression scores

The mean score for the total group for anxiety was 8.07 (sd 4.34). It was higher in those assigned female (8.69 (sd 4.32)) versus those assigned male at birth (7.54 (sd 4.31). The mean score for depression was 7.24 (sd 4.03), also higher in assigned females (7.48 (sd. 3.94)) than assigned males at birth (7.03 (sd. 4.11).

At T0 (before hormone treatment) 51.13% of participants scored 8 or over on the HADS-A subscale and in the case of the HADS-D subscale 47.75% of participants scored above 8 placing these participants within the categories of possible to probable presence of an anxiety or depression disorder. At T1 47.19% of participants scored 8 or above in the HADS-A subscale showing a reduction of 3.94% and in the HADS-D subscale 25.84% of people scored 8 or above showing a reduction of 21.91%.

Change in anxiety and depression scores between T0 and T1

There was a statistically significant reduction in mean scores of HADS- D from T0 to T1 (mean change difference -2.05, 95% CI, -2.72 - -1.38, p=0.00). This indicated a reduction in depression following 18 months of GAHT. There was also a reduction in the HADS-A score from T0 to T1, but this was not statistically significant (mean change difference -0.31, 95% CI, -0.97-0.36, p=0.37). The same findings (a significant reduction in HADS-D and a non-significant reduction in HADS-A) were found when comparing T0 and T1 according to sex assigned at birth (see Table 2).

Please insert Table 2 around here

Predictors of anxiety and depression change after hormone treatment Two multiple regressions with seven predictor variables were conducted to explore the predictors of change, from T0 to T1, in scores on HADS-D and HADS-A. The predictors for each were; ethnicity, employment status, relationship status, assigned sex at birth, age. MSPSS and AQ short at T0 were used as independent variables for both regressions.

The results for the first regression showed that overall the model was significant (F(7,152)=2.09, p=0.04) and explained 8.8% (R^2 =0.088) of the total variance in depression scores. The model also showed that mean MSPSS scores at T0 was the only significant predictor of HADS-D change between T0 and T1 (β =0.81, p=0.006). The second regression

showed that overall the model was significant (F(7,152)=2.09, p=0.048) and explained 8.8% (R^2 =0.088) of the total variance in anxiety scores. The model also showed mean AQ short scores at T0 was a significant predictor of HADS-A change between T0 and T1 (β =-0.069, p=0.034). The findings suggest that only levels of social support (MSPSS scores) and autistic spectrum traits (AQ scores) were able to predict changes in anxiety and depression following 18 months of GAHT. Having higher levels of social support (higher scores of MSPSS) predicted a reduction in depression scores following 18 months of GAHT (p=0.006) and having lower levels of autistic spectrums traits (lower AQ scores) predicted a reduction of anxiety symptoms following 18 months of GAHT (p=0.03), although this reduction was statistically non significant (See Table 3).

Please insert Table 3 around here

Discussion

This prospective longitudinal study aimed to explore whether 18 months of GAHT reduces symptoms of anxiety and depression in transgender people, whilst addressing the limitations of previous studies, by recruiting a large sample of participants within the same setting. The study found a significant reduction in symptoms of depression in transgender individuals after 18 months of starting GAHT, with a more than one fifth decrease in the number of participants who scores reflected a possible or probable depressive disorder. A statistical reduction of anxiety was not found. While reductions in depression, and to a lesser extent anxiety, were seen, a significant proportion of participants still present, posttreatment, with a possible or probable depressive disorder (25.84%) or anxiety disorder (47.19%). Data from previous studies in the field was used to compare these findings with the general population [13, 16]. Acknowledging that direct comparison can not be made, our study showed that the levels of possible and probable anxiety and depressive disorder after GAHT was still significantly higher than those reported in the general population (4.5% for possible or probable depressive disorder [16] and 34.5% for possible or probable anxiety disorder [13]. Whether these elevated levels will reduce further (when a longer use of GAHT +/- surgical interventions) needs to be explored. Thus, future longitudinal studies would benefit from following people for longer in order to track the longer-term impact of interventions.

These findings do confirm, once again, the high levels of possible anxiety and depressive disorders before GAHT and the benefit that this treatment brings. It highlights the need to facilitate the expedited use of GAHT to aid the reduction of poor mental health symptoms in the transgender population, when possible and appropriate. This conclusion supports the literature which has called for longitudinal studies such as this to replicate the findings from cross-sectional studies [15, 16, 22, 25, 38, 45]. The large reduction in depression, comparing to anxiety, may indicate that GAHT targets the dysphoria that many people attending transgender health services present with, which is manifested as depression (rather than anxiety). The fact that many transgender people still feel anxious after GAHT may be due to the victimisation, discrimination, and social rejection experienced by the transgender population [3, 5, 10, 53]. Unfortunately for some, these experiences do not necessarily stop

after initiation of GAHT and in some cases they may increase. Clinical and community services should take these findings into consideration and increase the support offered even after gender affirming medical treatment is over.

Importantly, this study also highlights how levels of pre-treatment social support is predictive of reduced risk post-treatment, since higher levels of social support prior to receiving GAHT significantly predicts a greater reduction in depression symptoms after 18 months of receiving GAHT. This indicates the importance of increasing social support in transgender people.

The study also found that those with higher levels of Autistic Spectrum Condition (ASC) traits prior to the receipt of GAHT had lower reductions in anxiety symptoms. This could indicate that those with greater ASC traits have higher anxiety symptoms even after post-treatment, or that higher ASC traits are simply reflective of difficulties in social interactions, as a result of being anxious, grounded in an individual's gender identity status. These findings corroborate cross-sectional studies which have shown that interpersonal interactions can have an impact upon transgender people's psychological wellbeing [9, 15, 23, 25, 27]. However, these results need to be interpreted with caution due to the lack of validity of the AQ short in this population.

With social support (from families, friends or significant others) being highlighted as a factor in depression it is important to consider approaches to aiding in the building of social support networks for transgender people accessing transgender services. Such support does not necessarily need to be based in clinical settings, since online and offline peer support and the work of tertiary services are available, less stigmatising, and have demonstrated a positive impact [29, 30, 54]. Resources such as Peer Support Workers (PSWs) and online peer-to-peer support may be a valuable tool to provide social support for transgender individuals awaiting and receiving GAHT. Additional support is recommended in particular to those with higher levels of ASC traits due to the lower reductions in anxiety symptoms found in this population. However, PSWs need to be appropriately trained in order to reduce any potential risks related to managing their own stress as well as power imbalances within peer and professional relationship [30].

Regarding expedited access to GAHT, there are practical issues that must be considered. [55, 56, 57]. Many countries either lack clinical services specializing in transgender health care or have significant waiting lists to access these services (13, 58-60]. These practical issues surrounding accessing treatment can lead to self-prescribing of GAHT, with 23% of individuals referred to transgender health clinics using GAHT prior to their first appointment, 70% of which was sourced online [61]. Self-prescribing without medical oversight presents its own risks, most notably a lack of specialized knowledge required to minimize health risks [61].

The issues surrounding access to GAHT may be compounded by how services are configured. The role of many mental health workers in transgender health services in the UK is seen as gatekeepers; focused on the assessment of transgender people with limited attention given to psychological support. A shift in roles from gatekeeping to tangible mental health support would allow for a partial addressing of the power imbalance between transgender people and mental health professionals. This would potentially allow for transgender people to feel more able to discuss issues without fear of rejection for treatment. This in turn would provide mental health professionals with the ability to focus more on supporting the mental health of those transgender people who need it. From the evidence provided by this study it appears that this would be most important for those with low levels of social support and those with autistic traits.

The strengths of this paper are the large sample and the naturalistic prospective longitudinal design within one transgender health service, which allows for within subject comparisons and so provides a highly valid insight into the impact of GAHT on depression and anxiety of treatment-seeking transgender people. It is important to add that this is a national NHS service, which offers gender affirming medical interventions and assessment free at the point of access to people from different geographical regions within the UK. This study is one of the few in the literature currently addressing the role of GAHT on mental health with such a methodology.

There are, however, limitations to the study. We must acknowledge that participants may downplay their symptoms of depression and anxiety pre- assessment, for fear of not being

treated. Consequently, participants may have attenuated their mental health symptoms due to concern of not being accepted for GAMT. This could indicate that the changes between T0 and T1 may be more significant than recorded. As the method of recruitment was through a transgender health clinic as part of the NHS the findings are only generalizable to treatment-seeking transgender individuals. A limitation may also be the response rate. Although this is low, it is in line with other clinical studies and it may be a reflection of the unfunded nature of the study. Another limitation is that the sample consisted of predominantly white participants, which may explain why ethnicity was not a predictive factor and the lack of control group where the intervention (GAHT) is not being offered (e.g., waiting list), but this will have its limitations too. A full randomised control study within this area will not be ethically possible. Studies using data from a clinical setting must also be aware of the context in which their data is being gathered.

In conclusion, this study shows that the mental health of transgender individuals improves following GAHT, particularly for those who reported high levels of social support prior to receiving GAHT. These results highlight the important mental health benefits of GAHT and emphasises and the need for interventions focused on developing social support for transgender individuals.

Authors contributions

JA conceptualised the study, participated in the study design. ZA SP JA WPB and BG handled data collection management. BG carried out the statistical analysis. ZA drafted the initial manuscript. SP drafted the initial methods and results section of the manuscript. All authors contributed to the critical review of the manuscript and approved the final version of the manuscript.

Acknowledgements

The authors thank all participants in the study for having given their time so generously.

Funding

The data analysis only was funded by the National Institute for Health Research (NIHR) Applied Research Collaboration East Midlands (ARC EM), NIHR Research for Patient Benefit (RfPB) and Nottinghamshire Healthcare NHS Foundation Trust.

Declaration of Interest

The authors have no other conflicts of interest relevant to this article to disclose. All procedures performed in studies involving human participants were in accordance with the ethical standards of the institutional and/or national research committee and with the 1964 Helsinki declaration and its later amendments or comparable ethical standards.

References

- Birkett M, Newcomb ME, Mustanski B. Does it get better? A longitudinal analysis of psychological distress and victimization in lesbian, gay, bisexual, transgender, and questioning youth. *J Adolesc Health*. 2015; 56(3): 280-285. doi: 10.1016/j.jadohealth.2014.10.275
- 2. Bouman WP, Arcelus J. The transgender handbook: A guide for transgender people, their families, and professionals. New York: Nova Publishers; 2017. ISBN: 978-1-53611-843-8

- 3. Breslow AS, Brewster ME, Velez BL, Wong S, Geiger E, Soderstrom, B. Resilience and collective action: Exploring buffers against minority stress for transgender individuals. *Psychol Sex Orientation Gender Div*. 2015; 2(3): 253-265. doi:10.1037/sgd0000117
- 4. Reisner SL, Vetters R, Leclerc M, Zaslow S, Wolfrum S, Shumer D, Mimiaga MJ. Mental health of transgender youth in care at an adolescent urban community health center: a matched retrospective cohort study. *J Adolesc Health*. 2015;56(3):274– 279. doi:10.1016/j.jadohealth.2014.10.264
- Carmel TC, Erickson-Schroth, L. Mental health and the transgender population. *J Psychosoc Nurs Ment Health Serv.* 2016; 54(12): 44-48.
 doi: 10.3928/02793695-20161208-09
- Claes L, Bouman WP, Witcomb G, Thurston M, Fernandez-Aranda F, Arcelus J. Non-suicidal self-injury in trans people: associations with psychological symptoms, victimization, interpersonal functioning, and perceived social support. *J Sex Med*. 2015; 12(1): 168-179. doi: 10.1111/jsm.12711
- 7. Dhejne C, Van Vlerken R, Heylens G, Arcelus J. Mental health and gender dysphoria: A review of the literature. *Int Rev Psychiatry*. 2016; 28(1): 44-57.
 doi: 10.3109/09540261.2015.1115753
- Seelman KL, Colon-Diaz MJP, LeCroix RH, Xavier-Brier M, Kattari L (2017). Transgender noninclusive healthcare and delaying care because of fear: connections to general health and mental health among transgender adults. *Transgender Health*. 2017; 2(1):17-28. doi: 10.1089/trgh.2016.0024.
- Simons L, Schrager SM, Clark LF, Belzer M, Olson J. Parental support and mental health among transgender adolescents. *J Adolesc Health*. 2013; 53(6): 791-793.
 doi: 10.1016/j.jadohealth.2013.07.019
- 10. Herman JL. Gendered Restrooms and Minority stress: The public regulation of gender and its impact on transgender people's lives. Los Angeles: The Williams Institute; 2013.
- Meyer IH. Prejudice, social stress, and mental health in lesbian, gay, and bisexual populations: conceptual issues and research evidence. *Psychol Bull*. 2003; 129(5): 674-697. doi: 10.1037/0033-2909.129.5.674

- 12. Meyer IH, Schwartz S, Frost DM. Social patterning of stress and coping: does disadvantaged social statuses confer more stress and fewer coping resources? *Soc Sci Med*. 2008; 67(3): 368-379. doi: 10.1016/j.socscimed.2008.03.012
- 13. Bouman WP, Claes L, Brewin N, Crawford JR, Millet N, Fernandez-Aranda Fernando, Arcelus J. Transgender and anxiety: a comparative study between transgender people and the general population. *Int J Transgenderism.* 2017; 18(1): 16-26. doi:10.1080/15532739.2016.1258352
- 14. Millet N, Longworth J, Arcelus J. Prevalence of anxiety symptoms and disorders in the transgender population: A systematic review of the literature. *Int J Transgenderism*. 2017, 18(1): 27-38. doi:10.1080/15532739.2016.1258353
- 15. Bouman WP, Davey A, Meyer C, Witcomb GL, Arcelus J. Predictors of psychological well-being among treatment seeking transgender individuals. *Sex Relats Ther.* 2016; 31(3): 359-375. doi:10.1080/14681994.2016.1184754
- 16. Witcomb GL, Bouman WP, Claes L, Brewin N, Crawford JR, Arcelus J. Levels of depression in transgender people and its predictors: results of a large matched control study with transgender people accessing clinical services. *J Affect Disord*. 2018; 235: 308-315. doi: 10.1016/j.jad.2018.02.051
- 17. Veale JF, Watson, RJ, Peter, T, Saewyc, EM. Mental health disparities among Canadian transgender youth. *J Adolesc Health*. 2017; 60(1): 44-49. doi: 10.1016/j.jadohealth.2016.09.014
- Marshall E, Claes L, Bouman WP, Witcomb GL, Arcelus J. Non-suicidal self-injury and suicidality in trans people: A systematic review of the literature. *Int Rev Psychiatry*. 2016; 28(1): 58-69. doi: 10.3109/09540261.2015.1073143
- Butler C, Joiner R, Bradley R, Bowles M, Bowes A, Russell C, Roberts V. Self-harm prevalence and ideation in a community sample of cis, trans and other youth. *Int J Transgenderism*.
 2019; 20(4): 447-458. doi: 10.1080/15532739.2019.1614130
- 20. Davey A, Arcelus J, Meyer C, Bouman WP. (2016). Self-injury among trans individuals in transition and matched controls: Prevalence and associated factors. *Health & Social Care in the Community*, 24, 485-494. doi: 10.1111/hsc.12239.
- Bergero-Miguel T, Garcia-Encinas MA, Villena-Jimena A, Perez-Costillas L, Sanchez-Alvarez N,
 de Diego-Otero Y, Guzman-Parra J. Gender Dysphoria and social anxiety: An exploratory study in Spain. *J Sex Med*. 2016; 13(8): 1270-1278.

doi: 10.1016/j.jsxm.2016.05.009

- 22. Meier CSL, Fitzgerald KM, Pardo ST, Babcock J. The effects of hormonal gender affirmation treatment on mental health in female-to-male transsexuals. *J Gay & Lesbian Mental Health*. 2011; *15*(3): 281–299. doi.org/10.1080/19359705.2011.581195
- 23. Nobili A, Glazebrook C, Bouman WP, Glidden D, Baron-Cohen S, Allison C, Smith P, Arcelus, J. Autistic traits in treatment-seeking transgender adults. *J Autism Dev Dis*. 2018; 48(12): 3984-3994. doi: 10.1007/s10803-018-3557-2
- 24. Nobili A, Glazebrook C. Bouman WP, Baron-Cohen S, Allison C, Arcelus J. The stability of autistic traits in transgender adults following cross-sex hormone treatment. *Int J Transgender Health*. 2020; in press.
- 25. Budge SL, Adelson JL, Howard KAS. Anxiety and depression in transgender individuals: the roles of transition status, loss, social support, and coping. *J Consult Clin Psychol*. 2013; 81(3): 545-557. doi: 10.1037/a0031774
- 26. Davey A, Bouman WP, Arcelus J, Meyer C. (2014). Social support and psychological wellbeing: a comparison of patients with gender dysphoria and matched controls. *J Sex Med*. 2014; 11(12): 2976–2985. doi: 10.1111/jsm.12681
- 27. Hoffman B. An overview of depression among transgender women. *Depress Res Treat*. 2014;2014:394283. doi: 10.1155/2014/394283.
- 28. Olson KR, Durwood L, DeMeules, M, McLaughlin KA. Mental health of transgender children who are supported in their identities. *Pediatrics*. 2016; 137(3): e20153223. doi: 10.1542/peds.2015-3223.
- 29. Naslund JA, Aschbrenner KA, Marsch LA, Bartels SJ. The future of mental health care: peerto-peer support and social media. *Epidemiol Psychiatr Sci*. 2016; 25(2): 113-122. doi: 10.1017/S2045796015001067
- 30. Repper J, Carter T (2011). A review of the literature on peer support in mental health services. *J Ment Health*. 2011; 20(4): 392-411. doi: 10.3109/09638237.2011.583947
- Jefferson K, Neilands TB, Sevelius J. Transgender women of color: discrimination and depression symptoms. *Ethn and Inequal Health Soc Care*. 2013; 6(4): 121–136. doi:10.1108/EIHSC-08-2013-0013
- 32. Dooley D, Catalano R, Wilson G. (1994). Depression and unemployment: panel findings from the Epidemiologic Catchment Area study. *Am J Community Psychol*. 1994; *22*(6): 745–765. doi: 10.1007/BF02521557

- 33. Leppel K. The labor force status of transgender men and women. *Int J Transgenderism*. 2016; 17(3-4): 155-164. doi:10.1080/15532739.2016.1236312
- 34. Mizock L, Mueser KT. Employment, mental health, internalized stigma, and coping with transphobia among transgender individuals. *Psychol Sex Orientation Gender Div*. 2014; 1(2): 146-158. doi:10.1037/sgd0000029
- 35. Mizock L, Woodrum TD, Riley J, Sotilleo EA, Yuen N, Ormerod AJ. Coping with transphobia in employment: strategies used by transgender and gender-diverse people in the United States. *Int J Transgenderism*. 2017; 18(3): 282-292. doi:10.1080/15532739.2017.1304313
- 36. Witcomb GL, Claes L, Bouman WP, Nixon E, Motmans J, Arcelus J. Experiences and psychological wellbeing outcomes associated with bullying in treatment-seeking transgender and gender-diverse youth. *LGBT Health*. 2019; 6(5): 216-226. doi: 10.1089/lgbt.2018.0179.
- 37. Arcelus J, Claes L, Witcomb GL, Marshall E, Bouman WP. Risk factors for non-suicidal self-injury among trans youth. *J Sex Med*. 2016; 13(3): 402-412.
 doi: 10.1016/j.jsxm.2016.01.003
- 38. Colizzi M, Costa R, Todarello O. Transsexual patients' psychiatric comorbidity and positive effect of cross-sex hormonal treatment on mental health: results from a longitudinal study. *Psychoneuroendocrinology*. 2014; 39(1): 65-73. doi: 10.1016/j.psyneuen.2013.09.029
- 39. Heylens G, Verroken C, De Cock S, T'Sjoen G, De Cuypere G. Effects of different steps in gender reassignment therapy on psychopathology: A prospective study of persons with a gender identity disorder. *J Sex Med*. 2014; 11(1): 119-126. doi: 10.1111/jsm.12363.
- 40. Bränström R, Pachankis JE. Reduction in mental health treatment utilization among transgender individuals after gender-affirming surgeries: a total population study. *Am J Psychiatry*. 2019; in press. doi: 10.1176/appi.ajp.2019.19010080
- 41. Zigmond AS, Snaith RP. The hospital anxiety and depression scale. *Acta Psychiatr Scand*. 1983;67(6):361-370. doi: 10.1111/j.1600-0447.1983.tb09716.x.
- 42. Bjelland I, Dahl AA, Haug TT, Neckelmann D. The validity of the hospital anxiety and depression scale. An updated literature review. *J Psychosom Res*. 2002; 52(2): 69-77. doi: 10.1016/s0022-3999(01)00296-3.
- 43. Gómez-Gil E, Zubiaurre-Elorza L, Esteva I, Guillamon A, Godás T, Cruz Almaraz M, Halperin I,
 Salamero M. Hormone-treated transsexual report less social distress, anxiety, and
 depression. *Psychoneuroendocrinology*. 2012; 37(5): 662–670.

doi: 10.1016/j.psyneuen.2011.08.010.

- 44. Zimet GD, Powell SS, Farley GK, Werkman S, Berkoff KA. Psychometric characteristics of the Multidimensional Scale of Perceived Social Support. *J Pers Assess*. 1990; 55: 610-617. doi: 10.1080/00223891.1990.9674095
- 45. Bouman WP, Claes L, Marshall E, Pinner GT, Longworth J, Maddox V, Witcomb G, Jimenez-Murcia S, Fernandez-Aranda F, Arcelus J. Socio-demographic variables, clinical features and the role of pre-assessment cross-sex hormones in older trans people. *J Sex Med*. 2016; 13(4): 711-719. doi: 10.1016/j.jsxm.2016.01.009.
- 46. Hoekstra RA, Vinkhuyzen AAE, Wheelwright S, Bartels M, Boomsma DI, Baron-Cohen S, Posthuma D, van der Sluis S. The construction and validation of an abridged version of the Autism-Spectrum Quotient (AQ-Short). *J Autism Dev Disord*. 2011; 41(5): 589–596. doi:10.1007/s10803-010-1073-0
- 47. Woodbury-Smith MR, Robinson J, Wheelwright S, Baron-Cohen S. Screening adults for Asperger Syndrome using the AQ: a preliminary study of its diagnostic validity in clinical practice. J Autism Dev Disord. 2005; 35(3): 331–335. https://doi.org/10.1007/s10803-005-3300-7
- 48. Baron-Cohen S, Wheelwright S, Skinner R, Martin J, Clubley E. The Autism-Spectrum Quotient (AQ): Evidence from Asperger Syndrome/High-Functioning Autism, Males and Females, Scientists and Mathematicians. *J Autism Dev Disord*. 2001; 31(1): 5–17. doi:10.1023/A:1005653411471
- 49. Pasterski V, Gilligan L, Curtis R. Traits of autism spectrum disorders in adults with gender dysphoria. *Arch Sex Behavior*. 2014; 43(2):387-393. doi: 10.1007/s10508-013-0154-5.
- 50. StataCorp. 2019. Stata Statistical Software: Release 16. College Station, TX: StataCorp LLC.
- 51. Allison PD. Comparing logit and probit coefficients across groups. *Sociol Methods & Res*. 1999; 28(2): 186-208. doi:10.1177/0049124199028002003
- 52. Health Research Authority (HRA). Guidance for NHS Research Studies. London: Health Research Authority; 2013.
- 53. Fuller KA, Riggs DW. Family support and discrimination and their relationship to psychological distress and resilience amongst transgender people. A systematic review. *Int J Transgenderism.* 2018; 19(4): 379-388. doi: 10.1080/15532739.2018.1500966

- 54. Bockting WO, Miner MH, Swinburne Romine RE, Hamilton A, Coleman E. Stigma, mental health, and resilience in an online sample of the US transgender population. *Am J Public Health*. 2013; 103(5): 943–951. doi:10.2105/AJPH.2013.301241
- 55. Bouman WP, de Vries ALC, T'Sjoen G. Gender Dysphoria and Gender Incongruence: An evolving inter-disciplinary field. *Int Rev Psychiatry*. 2016; 28(1): 1-4. doi: 10.3109/09540261.2016.1125740
- 56. Motmans J, Nieder TO, Bouman WP. Transforming the paradigm of nonbinary transgender health: A field in transition. *Int J Transgenderism*. 2019; 20(2+3): 118-124. doi: 10.1080/15532739.2019.1640514
- 57. Richards C, Arcelus J, Barrett J, Bouman WP, Lenihan P, Lorimer S, Murjan S, Seal L. Trans is not a disorder – but should still receive funding. *Sex Relats Ther*. 2015; 30(3): 309-313. doi:10.1080/14681994.2015.1054110
- Bouman WP, Richards C. Diagnostic and treatment issues for people with gender dysphoria in the United Kingdom. *Sex Relats Ther*. 2013; 28(3): 165-171. doi:10.1080/14681994.2013.819222
- 59. Davies A, Bouman WP, Richards C, Barrett J, Ahmad A, Baker K, Lenihan P, Lorimer S, Mepham N, Murjan S, Robbins-Cherry S, Seal L, Stradins L. (2013). Patient satisfaction with gender identity clinic services in the United Kingdom. *Sex Relats Ther*. 2013; 28(4): 400-418. doi:10.1080/14681994.2013.834321
- 60. Winter S, Diamond M, Green J, Karasic D, Reed T, Whittle S, Wylie K. (2016). Transgender people: health at the margins of society. *Lancet*. 2016; 388(10042): 390-400. doi:10.1016/S0140-6736(16)00683-8
- 61. Mepham N, Bouman WP, Arcelus J, Hayter M, Wylie KR. People with gender dysphoria who self-prescribe cross-sex hormones: prevalence, sources, and side effects knowledge. *J Sex Med*. 2014; 11(12): 2995-3001. doi: 10.1111/jsm.12691

Table 1.

Socio-demographic characteristics (n,%) and mean (SD) scores on HADS, MSPSS and AQ-short

Score of all respond	lers assigned mal	e and female at birth
----------------------	-------------------	-----------------------

Responders			Assigned male at birth (2)			Assigned female at birth (1)	
	n = 178		n = 95		n = 83		
	n %		n %		n %		
Age, median (range) years	23 (17-74)		28 (17-79)		21 (17-64)		
Ethnic origin n (%)							
White	167	(94)	92	(97)	75	(91)	
Other	11	(6)	3	(3)	8	(9)	
Not known	-	-	-	-	-	-	
Employment status n (%)							
Employed (1)	75	(41)	45	(47)	28	(33)	
Student (2)	58	(32)	20	(21)	37	(44)	
Housewife/househusband(4)	-	-	-	-	-	-	
Voluntary work (3)	7	(4)	5	(5)	2	(2)	
Retired (6)	5	(3)	6	(6)	-	-	
Disabled (5)	7	(5)	5	(5)	3	(3)	
Unemployed (0)	26	(14)	14	(14)	12	(14)	
Not known (9)	1	(1)	-	-	1	(1)	
Civil status n (%)							
Single (1)	120	(69)	55	(57)	65	(78)	
Married (2)	15	(7)	12	(12)	1	(1)	
Civil partner (3)	5	(3)	-	-	5	(6)	
In a relationship (7)	4	(3)	1	(1)	4	(4)	
Divorced/separated (4)	20	(11)	18	(18)	2	(2)	
Widowed (5)	2	(2)	2	(2)	1	(1)	
Other (6)	-	-	-	-	-	-	
Not known (9+Blanks)	12	(7)	7	(7)	5	(6)	
HADS- D	7.24 (4.03)		7.03 (4.11)		7.48 (3.94)		
HADS-A	8.07 (4.34)		7.54 (4.31)		8.69 (4.32)		
MSPSS	4.85(1.29)		4.64(1.35)		5.1(1.16)		
AQ	64.77(11.86)		62.83(11.94)		66.97(11.44)		

Table 2. Means (SD) of HADS-D and HADS-A scores of responders at T0 and T1

_									
All responders			Assigned male at birth			Assigned female at birth			
	Ν	Mean	Mean change	n	Mean (sd)	Mean change from	n	Mean (sd)	Mean change from
		(sd)	from T0 to T1			T0 to T1 (95% CI),p-			T0 to T1 (95% CI),p-
			(95% CI),p-value			value			value
HADS	i-D								
то	178	7.24	-2.05	95	7.03 (4.11)	-1.91	83	7.48 (3.94)	-2.21
		(4.03)	(-2.72 – -1.38),			(-2.801.01)			(-3.23 – -1.20)
T1	178	5.19	P= 0.00	95	5.13 (3.92)	p =0.00	83	5.26 (3.52)	p = 0.00
		(3.73)							
HADS	i-A								
т0	178	8.07	-0.31	95	7.54 (4.31)	-1.16	83	8.69 (4.32)	-0.55
		(4.34)	(-0.97 – 0.36),			(-1.50-0.39)			(-0.97 – 0.92) p=0.97
T1	178	7.77	p =0.37	95	6.98 (3.96)	p =0.25	83	8.66 (3.65)	
		(3.90)							

Accepted

Ì

Table 3. Predictive role of ethnicity, employment status, relationship status, assigned gender atbirth, age, MSPSS and AQ short for change in HADS-D and HADS-A from T0 to T1.

HADS-D			HADS-A			
	Coef.	95% CI	р	Coef.	95% CI	Р
Ethnicity (grouped as White and all other at T0)	-0.37	-3.19 – 2.45	0.794	-0.67	-3.62 - 2.28	0.652
Employment (grouped as unemployed and disabled and all other at T0)	-0.97	-2.75-0.81	0.284	-0.51	-2.37-1.35	0.591
Relationship (grouped as single, widowed, divorced/separated and other at TO and all other)	0.31	-1.82 – 2.43	0.776	-0.61	-2.83 – 1.62	0.590
Assigned sex at T0	0.36	-1.21– 1.93	0.651	-0.17	-1.81 - 1.47	0.841
Age at TO	0.06	0.50 – 0.61	0.843	-0.03	-0.08 - 0.03	0.409
Mean MSPSS at TO	0.81	0.24– 1.39	0.006	0.56	-0.04 - 1.16	0.065
Mean AQ short at TO	-0.04	-0.096 – 0.025	0.250	-0.07	-0.13 – -0.05	0.034
Constant	-1.93	-4.76 – 0.90	0.179	0.41	-2.34 - 3.16	0.787

HORMONE TREATMENT REDUCES SYMPTOMS OF POOR MENTAL HEALTH IN TRANS PEOPLE

Trans people seeking gender affirming hormone treatment (GAHT) who have not yet received it often experience higher levels of depression and anxiety.



* In the scales used to assess the symptoms of depression (HADS-D) and anxiety (HADS-A) experienced by participants, a score of 8 or over indicated a possible or probable depression or anxiety disorder.



Gender affirming hormone treatment (GAHT) has a significant positive impact on the mental health of treatment seeking transgender people

Those with strong social support before GAHT are more likely to experience reductions in levels of depression